Author’s response to reviews

Title: Impact of three different plate colours on short term satiety and energy intake: a randomised controlled trial

Authors:

Asli akyol (asli.akyol@hacettepe.edu.tr)
Aylin Ayaz (baylin@hacettepe.edu.tr)
Elif Inan-Eroglu (elif.inan@hacettepe.edu.tr)
Cansu Cetin (cetin.cansu@hacettepe.edu.tr)
Gulhan Samur (gsamur@hacettepe.edu.tr)

Version: 3 Date: 05 Feb 2018

Author’s response to reviews:

We would like to thank for your careful and constructive reviews. The corrections in red address proof reading (edited by American Journal Experts) and yellow address corrections for reviewer comments.